Author's response to reviews

Title: Atherosclerosis, inflammation and lipoprotein glomerulopathy in kidneys of apoE-/-/LDL-/- double knockout mice

Authors:

Alexander C Langheinrich (alexander.langheinrich@radiol.med.uni-giessen.de)
Marian Kampschulte (marian.kampschulte@radiol.med.uni-giessen.de)
Franziska Scheiter (franziska_scheiter@web.de)
Christian Dierkes (christian.dierkes@ukmuenster.de)
Philip Stieger (philip.stieger@innere.med.uni-giessen.de)
Rainer M Bohle (Rainer.Bohle@uniklinikum-saarland.de)
Wolfgang Weidner (wolfgang.weidner@chiru.med.uni-giessen.de)

Version: 4 Date: 19 August 2010

Author's response to reviews: see over
Dear Editor,

This is a re-submission of our accepted manuscript entitled

“Atherosclerosis, Inflammation and Lipoprotein Glomerulopathy in Kidneys of ApoE-/-/LDL-/- double knockout mice”.

The authors thank you and the reviewers for the comments and recommendations. We answered concerns on a point-by-point basis.

We hope that the present version of the manuscript is now acceptable for publication in your journal.

Yours sincerely,

Alexander C. Langheinrich
Formatting changes to be made:
--------------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.

AUTHORS: As indicated at the reference guide (http://www.biomedcentral.com/bmcgenet/ifora/#references) the current version of the manuscript matches exactly the style of the manuscript. As mentioned above, the preferred style is: 'First name Initial Last name' (e.g. Joe F Bloggs). By checking recently published manuscripts and the reference guide, the recommended list of authors appears as:


Please let us know if we should re-organize the author list.

Figures: Please ensure that each figure has a short overall title. Each figure should have a single title- figures can have multiple parts and corresponding legends, but should have a single composite/overall title.

AUTHORS: A short overall title has been implemented in the revised version of the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the city in the affiliation details.

AUTHORS: The city has been implemented in the affiliation details.

Text: Please ensure that all text is displayed in black throughout the manuscript.

AUTHORS: The text is now displayed in black.

References: please provide the full citation details for reference 12. You can view the BMC reference guide at the following link

http://www.biomedcentral.com/bmcgenet/ifora/#references

AUTHORS: Full citation details for reference 12 are now provided.